ORIGINAL RESEARCH ARTICLE





Blinded Withdrawal of Long-Term Randomized Treatment With Empagliflozin or Placebo in Patients With Heart Failure

Milton Packer[®], MD; Javed Butler[®], MD, MPH, MBA; Cordula Zeller[®], Dipl. Math.; Stuart J. Pocock[®], PhD; Martina Brueckmann[®], MD; João Pedro Ferreira[®], MD; Gerasimos Filippatos[®], MD; Muhammad Shariq Usman, MD; Faiez Zannad[®], MD, PhD; Stefan D. Anker[®], MD, PhD

BACKGROUND: It is not known whether the benefits of sodium-glucose cotransporter 2 inhibitors in heart failure persist after years of therapy.

METHODS: In the EMPEROR-Reduced (Empagliflozin Outcome Trials in Chronic Heart Failure With Reduced Ejection Fraction) and EMPEROR-Preserved (Empagliflozin Outcome Trials in Chronic Heart Failure With Preserved Ejection Fraction) trials, patients with heart failure were randomly assigned (double-blind) to placebo or empagliflozin 10 mg/day for a median of 16 and 26 months, respectively. At the end of the trials, 6799 patients (placebo 3381, empagliflozin 3418) were prospectively withdrawn from treatment in a blinded manner, and, of these, 3981 patients (placebo 2020, empagliflozin 1961) underwent prespecified in-person assessments after ≈30 days off treatment.

RESULTS: From 90 days from the start of closeout to the end of double-blind treatment, the annualized risk of cardiovascular death or hospitalization for heart failure was lower in empagliflozin-treated patients than in placebo-treated patients (10.7 [95% CI, 9.0–12.6] versus 13.5 [95% CI, 11.5–15.6] events per 100 patient-years, respectively; hazard ratio 0.76 [95% CI, 0.60–0.96]). When the study drugs were withdrawn for \approx 30 days, the annualized risk of cardiovascular death or hospitalization for heart failure increased in patients withdrawn from empagliflozin but not in those withdrawn from placebo (17.0 [95% CI, 12.6–22.1] versus 14.1 [95% CI, 10.1–18.8] events per 100 patient-years for empagliflozin and placebo, respectively). The hazard ratio for the change in risk in the patients withdrawn from empagliflozin was 1.75 (95% CI, 1.20–2.54), *P*=0.0034, whereas the change in the risk in patients withdrawn from placebo was not significant (hazard ratio 1.12 [95% CI, 0.76–1.66]); time period-by-treatment interaction, *P*=0.068. After withdrawal, the Kansas City Cardiomyopathy Questionnaire Clinical Summary Score declined by 1.6±0.4 in patients withdrawn from empagliflozin versus placebo (*P*<0.0001). Furthermore, withdrawal of empagliflozin was accompanied by increases in fasting glucose, body weight, systolic blood pressure, estimated glomerular filtration rate, N-terminal pro-hormone B-type natriuretic peptide, uric acid, and serum bicarbonate and decreases in hemoglobin and hematocrit (all *P*<0.01). These physiological and laboratory changes were the inverse of the effects of the drug seen at the start of the trials during the initiation of treatment (\approx 1–3 years earlier) in the same cohort of patients.

CONCLUSIONS: These observations demonstrate a persistent effect of empagliflozin in patients with heart failure even after years of treatment, which dissipated rapidly after withdrawal of the drug.

REGISTRATION: URL: https://www.clinicaltrials.gov; Unique identifiers: NCT03057977 and NCT03057951.

Key Words: empagliflozin = heart failure = sodium-glucose transporter 2 inhibitors = substance withdrawal syndrome

Correspondence to: Milton Packer, MD, Baylor Heart and Vascular Institute, Baylor University Medical Center, 621 North Hall St, Dallas, TX 75226. Email milton.packer@baylorhealth.edu

Supplemental Material, the podcast, and transcript are available with this article at https://www.ahajournals.org/doi/suppl/10.1161/CIRCULATIONAHA.123.065748. Continuing medical education (CME) credit is available for this article. Go to http://cme.ahajournals.org to take the quiz.

© 2023 The Authors. *Circulation* is published on behalf of the American Heart Association, Inc., by Wolters Kluwer Health, Inc. This is an open access article under the terms of the Creative Commons Attribution Non-Commercial-NoDerivs License, which permits use, distribution, and reproduction in any medium, provided that the original work is properly cited, the use is noncommercial, and no modifications or adaptations are made.

This work was presented as an abstract at ESC Congress, August 25-28, 2023.

For Sources of Funding and Disclosures, see page 1021.

Circulation is available at www.ahajournals.org/journal/circ

ORIGINAL RESEARCH Article

Clinical Perspective

What Is New?

- At the end of the EMPEROR-Reduced (Empagliflozin Outcome Trial in Patients With Chronic Heart Failure With Reduced Ejection Fraction) and EMPEROR-Preserved (Empagliflozin Outcome Trial in Patients With Chronic Heart Failure With Preserved Ejection Fraction) trials, 6799 patients with heart failure who had been randomly assigned (double-blind) to placebo or empagliflozin 10 mg/ day for 1 to 3 years were prospectively withdrawn (by protocol) from treatment for ≈30 days, with blinded prespecified clinical, physiological, and laboratory assessments immediately before and after treatment withdrawal.
- When compared with patients withdrawn from placebo, patients withdrawn from treatment with empagliflozin showed increased risk of major heart failure events and worsening health status, along with physiological and laboratory changes that were the inverse of those seen during the initiation of treatment 1 to 3 years earlier.

What Are the Clinical Implications?

- It is not known whether the benefits of foundational drugs persist after years of treatment, because formal withdrawal trials have not been performed.
- We demonstrated a persistent effect of empagliflozin in patients with heart failure even after years of treatment, which dissipated rapidly after withdrawal of the drug. These findings indicate that tolerance does not develop during long treatment and that abrupt cessation of empagliflozin even for short periods of time may have serious consequences.
- The proximal tubular effect of SGLT2 inhibitors elicits counterregulatory downstream antinatriuretic and antiaquaretic mechanisms, and abrupt withdrawal may allow these mechanisms to produce rebound phenomena.

Inical trials have established the efficacy of inhibitors of the renin-angiotensin system, β -blockers, mineralocorticoid receptor antagonists, neprilysin inhibitors, and sodium-glucose cotransporter 2 (SGLT2) inhibitors in patients with heart failure.^{1,2} Recommended practice is to maintain these drugs indefinitely in patients who can tolerate therapy, on the basis of the assumption that pharmacological tolerance does not develop during prolonged administration; yet, tolerance is common with many drugs for heart failure.^{3–6} Absence of a change in clinical and physiological variables after the withdrawal of a drug would support the development of tolerance.

Some observational studies have compared the clinical course of patients in whom treatment was stopped for a specific clinical reason with the course of patients who were maintained on therapy, typically showing that

Nonstandard Abbreviations and Acronyms

eGFR	estimated glomerular filtration rate
EMPEROR	Empagliflozin Outcome Trials in Chronic Heart Failure
EMPEROR-Preserved	Empagliflozin Outcome Trials in Chronic Heart Failure With Preserved Ejection Fraction
EMPEROR-Reduced	Empagliflozin Outcome Trials in Chronic Heart Failure With Reduced Ejection Fraction
KCCQ-CSS	Kansas City Cardiomy- opathy Questionnaire Clinical Summary Score
NT-proBNP	N-terminal pro-hormone B-type natriuretic peptide
SGLT2	sodium-glucose cotrans- porter 2
TRED-HF	A Pilot Feasibility Study in Recovered Heart Failure

patients who discontinued treatment fared worse than those who continued on treatment.7-11 However, these studies are substantially confounded, because the reasons for stopping treatment (rather than its discontinuation) drives the unfavorable prognosis of these patients. The durability of a drug effect can only be assessed in a protocol-specified manner, either in (1) a randomized blinded trial of placebo versus active treatment in a cohort of patients who received long-term therapy or (2) a blinded withdrawal trial of patients who have received long-term randomized treatment with placebo or active therapy. Although both approaches have been used in the past,^{12,13} large-scale withdrawal trials have not been performed after long-term treatment with any of the currently recommended foundational drugs for heart failure. The TRED-HF trial (Pilot Feasibility Study in Recovered Heart Failure)¹⁴ performed an open-label withdrawal of several drugs, but it studied only 51 asymptomatic patients with a nonischemic cardiomyopathy with a recovered ejection fraction, and it primarily assessed biomarkers of ventricular function.

In contrast with other landmark trials in heart failure, the EMPEROR (Empagliflozin Outcome Trials in Chronic Heart Failure) program with empagliflozin was designed so that all patients completing double-blind therapy underwent a protocol-specified withdrawal of their study medication for \approx 30 days after receiving \approx 1 to 3 years of treatment, thus allowing patients withdrawn from the SGLT2 inhibitor to be compared with patients withdrawn from placebo. Withdrawal measurements were made

METHODS

The EMPEROR Program consisted of 2 simultaneously conducted, multicenter double-blind randomized parallel-group placebo-controlled trials, which evaluated patients with chronic heart failure with an ejection fraction \leq 40% (enrolled in EMPEROR-Reduced [Empagliflozin Outcome Trials in Chronic Heart Failure With Reduced Ejection Fraction]) or >40% (enrolled in EMPEROR-Preserved [Empagliflozin Outcome Trials in Chronic Heart Failure With Preserved Ejection Fraction]).^{21,22} The 2 trials were performed by the same Executive Committee, and major clinical events were adjudicated by the same end point adjudication committee. It was prespecified that patient-level data would be combined for relevant analyses.²³

Data Sharing

To ensure independent interpretation of clinical study results and enable authors to fulfil their role and obligations under the International Committee of Medical Journal Editors criteria, Boehringer Ingelheim grants all external authors access to relevant clinical study data. In adherence with the Boehringer Ingelheim Policy on Transparency and Publication of Clinical Study Data, scientific and medical researchers can request access to clinical study data after publication of the primary article and secondary analyses in peer-reviewed journals and regulatory and reimbursement activities are completed, normally within 1 year after the marketing application has been granted by major Regulatory Authorities. Researchers should use the https://vivli.org/ link to request access to study data and visit https://www.mystudywindow.com/msw/datasharing for further information.

Patient Population

The design and primary results of both trials have been previously published.^{21,22} Participants were men or women, ≥ 18 years of age with New York Heart Association functional class II to IV heart failure for ≥3 months. Patients were required to have an elevated N-terminal pro-hormone B-type natriuretic peptide (NT-proBNP) levels (ie, >300 pg/mL if the ejection fraction was >40%; ≥2500 pg/mL if the ejection fraction was 36%-40%; ≥ 1000 pg/mL if the ejection fraction was 31%-35%; and \geq 600 pg/mL if the ejection fraction was \leq 30% or if patients had been hospitalized for heart failure within 12 months). If patients had atrial fibrillation, these thresholds were doubled in EMPEROR-Reduced and tripled in EMPEROR-Preserved. In both trials, patients were randomly assigned to receive either placebo or empagliflozin (10 mg once daily), in addition to recommended therapy. Eligible patients were randomly assigned to placebo or empagliflozin 10 mg daily, which were added to all recommended therapy for heart failure, for the duration of double-blind follow-up (median 16 months in EMPEROR-Reduced and 26 months in EMPEROR-Preserved). Randomization was stratified by geographical region, diabetes status, and estimated glomerular filtration rate (eGFR, <60 or \geq 60 mL·min⁻¹·1.73 m⁻²) in EMPEROR-Reduced and by the same variables and also by left ventricular ejection fraction (<50% or \geq 50%) in EMPEROR-Preserved. The protocol was approved by the local ethical committee at each of the participating sites, and all patients gave written informed consent.

Withdrawal of Randomized Double-Blind Treatment

As specified in the original protocol for both trials, all surviving patients who completed the double-blind active treatment phase as planned were to stop taking their study drugs and return ≈30 days later for a repeat clinical and laboratory assessment. Data on the occurrence of heart failure events before and after the cessation of treatment were collected either in person or by telephone in 6799 patients, which represented >93% of patients who were still taking their study medications 90 days before the start of the closeout period. Furthermore, more than half of the patients who stopped double-blind treatment as planned returned for a formal 30-day posttreatment study visit for the assessment of health status (by the Kansas City Cardiomyopathy Questionnaire Clinical Summary Score [KCCQ-CSS]), physical examination and laboratory testing. This in-person 30-day posttreatment visit occurred less frequently in the EMPEROR-Reduced trial (compared with the EMPEROR-Preserved trial) due to the restriction on patientinvestigator interactions during the height of the COVID-19 pandemic.

Clinical and Laboratory Assessments

The principal clinical assessments were (1) the risk of cardiovascular death and heart failure hospitalization during the 30-day off-treatment period, compared with the period starting 90 days before the start of the closeout of each trial to the planned end of double-blind therapy; and (2) the KCCQ-CSS, assessed during the last double-blind visit and during the 30-day posttreatment visit. In addition, the following physical examination and laboratory evaluations were performed at the last double-blind on-treatment visit and at the time of the 30-day off-treatment follow-up visit: (1) systolic blood pressure; (2) body weight; (3) hemoglobin and hematocrit; (4) eGFR; (5) NT-proBNP; (6) uric acid; (7) fasting glucose; (8) serum albumin; and (9) serum sodium, potassium, and bicarbonate. To provide context for these observations, changes in KCCQ-CSS and laboratory tests (except for fasting glucose) were examined after 4 and 12 weeks (for laboratory tests) and after 12 weeks (for KCCQ-CSS) after the initiation of doubleblind randomized therapy in the same patients who had valid measurements during the withdrawal period.

Statistical Analysis

The clinical features of the patients who participated and did not participate in the withdrawal study were compared using the *t*-test for continuous variables and χ^2 tests for categorical variables. Changes in laboratory assessments and KCCQ at on-treatment visits at weeks 4, 12, and last-value-on-treatment and measurements 30 days after planned treatment discontinuation were evaluated by a mixed model for repeated measures.

The mixed model for repeated measures compared the treatment effects at the withdrawal versus effects at last-value-ontreatment, and, to evaluate the possibility of a mirroring effect during withdrawal compared with the initiation of treatment, all mixed model for repeated measures models were based on the cohort of ≈4000 patients who contributed a valid withdrawal measurement after planned withdrawal (ie, measurements during initiation were performed in the same cohort who had measurements during withdrawal). We focused on between-group differences in the change from baseline at week 4 (or week 12 for KCCQ-CSS), and between-group differences at the end of withdrawal compared with the last-value-on-treatment, as well. Measurements after drug withdrawal were considered valid if made 23 to 45 days after the planned cessation of treatment, but patients who started open-label therapy with a SGLT2 inhibitor during the withdrawal period were not included in the analysis. The mixed model for repeated measures models were adjusted for baseline covariates of sex, age, study, diabetes status, eGFR, left ventricular ejection fraction and region, and the-baseline-for-the-evaluated-parameter-by-visit interaction, and they included a covariate for the time of recruitment, thus reflecting the different possible follow-up visits in this eventdriven trial.

Between-group differences in the risk of the composite of cardiovascular death or heart failure hospitalization were assessed during 2 periods: (1) from 90 days before the start of the closeout period up to the planned end of double-blind treatment and (2) from the end of double-blind treatment to the end of the withdrawal phase (which was planned for 30±7 days but included data up to 45 days after treatment discontinuation, so as to be consistent with the prespecified cutoff for valid postwithdrawal assessments). The periods were compared on the basis of the patients at risk during each time frame. The hazard ratios and 95% CIs were calculated using a Cox regression model, which was adjusted for the prespecified baseline covariates of age, sex, geographical region, diabetes, left ventricular ejection fraction, eGFR, and trial (EMPEROR-Reduced or EMPEROR-Preserved). In addition, we adjusted for baseline high-sensitivity cardiac troponin T and N-terminal pro-B type natriuretic peptide (NT-proBNP), the most important predictors of the primary outcome in the EMPEROR trial program,24,25 and for the use of foundational drugs for heart failure, as well (ie, mineralocorticoid receptor antagonist and sacubitril/valsartan) at time of the start of the 2 study periods. The assumption of proportional hazards was tested, and no violations were observed. Differences in hazard ratios between the treatment groups and across the 2 time periods were evaluated by a period-by-treatment interaction term in the model. Spearman correlation coefficient ρ was used to evaluate correlation of changes among hematocrit and NT-proBNP, body weight, and albumin within each treatment group in each of the 2 time periods.

All analyses were performed using SAS, version 9.4 (SAS Institute). All reported P values are 2 sided, with P<0.05 considered to be statistically significant.

RESULTS

Of the 3730 patients randomly assigned into the EM-PEROR-Reduced trial, 2925 patients (1446 placebo and 1479 empagliflozin) survived and were still taking their study medications until 90 days before the start of the closeout period. Because of variations in the start of closeout between sites and patients, the median duration of the end of double-blind phase was 110 days (interquartile range, 99-135 days). Because of COV-ID-19 restrictions, only 934 patients (25.0% of the total number of randomly assigned patients and 34.2% of the total number of eligible patients who completed doubleblind treatment) attended a formal 30-day off-treatment in-person study visit and provided valid postwithdrawal evaluation of KCCQ-CSS and laboratory tests. The median time from the last double-blind assessment to the off-treatment assessment was 29 days (interquartile range, 28–35 days). Of the 5988 patients randomly assigned into the EMPEROR-Preserved trial, 4368 patients (2177 placebo and 2191 empagliflozin) survived and were still taking their study medications until 90 days before the start of the closeout period. Because of variations in the start of closeout between sites, the median duration of the end of double-blind phase was 129 days (interquartile range, 120-142 days). Of these, 3047 patients (50.9% of the total number of randomly assigned patients and 74.4% of the total number of eligible patients who completed double-blind treatment) underwent a formal 30-day off-treatment in-person study visit and provided valid postwithdrawal evaluation of KCCQ and laboratory tests. The median time from the last doubleblind assessment to the off-treatment assessment was 28 days (interquartile range, 28-33 days).

When the 2 trials were combined (Figure S1), of 9718 patients who were randomly assigned to double-blind therapy, 7293 were receiving double-blind treatment 90 days before the start of the closeout period (placebo 3623, empagliflozin 3670). Of these, 6799 patients (placebo 3381, empagliflozin 3418) completed double-blind treatment and provided data after withdrawal, and, of these, 3981 patients (placebo 2020, empagliflozin 1961) completed the ≈30-day withdrawal period and underwent valid in-person off-treatment clinical and laboratory assessments. A total of 92 patients who started open-label therapy with a SGLT2 inhibitor during the withdrawal period were not included in the analysis.

Table S1 and Table S2 show the baseline prerandomization characteristics in the patients who contributed or did not contribute data to the evaluation of end-oftrial heart failure events during the withdrawal period or contributed or did not contribute data to the evaluation of paired assessments of KCCQ-CSS and laboratory tests at the end of double-blind treatment and \approx 30 days after withdrawal. In both instances, the patients who survived to participate in the withdrawal phase of the trials had less severe heart failure (as reflected by New York Heart Association functional class, ejection fraction, NT-proBNP and troponin levels, and recency of a heart failure hospitalization) and had fewer or less severe comorbidities (as reflected by diabetes and eGFR), compared with the full cohort of originally randomly assigned patients. In both Table S1 and Table S2, the placebo and empagliflozin groups were balanced for all reported characteristics.

At the end of double-blind treatment immediately before the 30-day withdrawal period, compared with the placebo group, the patients in the empagliflozin group had higher KCCQ-CSS scores, hemoglobin, and hematocrit, and lower systolic blood pressure, NT-proBNP, serum uric acid, and serum bicarbonate levels (all P<0.05; Table 1).

Clinical Assessments

During the 90 days before the start of closeout until the end of double-blind treatment, the annualized risk of cardiovascular death or hospitalization for heart failure was lower in the empagliflozin group than in the placebo group (10.7 [95% CI, 9.0-12.6] versus 13.5 [95% CI, 11.5-15.6] events per 100 patient-years, respectively; hazard ratio 0.76 [95% CI, 0.60-0.96]). However, when the study drugs were withdrawn for ≈30 days, the annualized risk of major heart failure events increased in patients withdrawn from empagliflozin but not in patients withdrawn from placebo (17.0 [95% Cl, 12.6-22.1] versus 14.1 [95% CI, 10.1-18.8] events per 100 patient-years for empagliflozin and placebo, respectively; between-group hazard ratio 1.18 [95% CI, 0.78-1.80]). The change in the annualized risk in the patients withdrawn from empagliflozin (from the on-treatment period to the off-treatment period) was significant (HR 1.75 [95% CI, 1.20-2.54], P=0.0034), whereas the change in the annualized risk in patients withdrawn from placebo was not significant (HR 1.12 [95% CI, 0.76-1.66]); P=0.068) for the time-period-by-treatment interaction, which compared between-group differences before and after planned end of treatment (Figure 1A). These findings were driven primarily by hospitalizations for heart failure (Table 2). The number of events during the withdrawal phase was small (49 versus 40 events in the empagliflozin and placebo groups), as shown in Table 2. Time-to-event plots for the prewithdrawal and postwithdrawal periods are shown in Figure S1. The patterns were similar when the 2 trials were analyzed separately (Table S3).

When double-blind therapy was withdrawn, we observed a 1.6 \pm 0.4 (adjusted mean \pm SE) greater decline in the KCCQ-CSS score at 30 days in the patients who had been withdrawn from treatment with empagliflozin compared with those withdrawn from treatment with placebo (*P*<0.0001; Figure 1B). The magnitude of this decline was similar to the magnitude of improvement that had been observed in these same patients after the first postrandomization assessment of KCCQ-CSS performed at the 12-week study visit.^{16,17}

Table 1. Clinical Characteristics of Patients in Placebo and Empagliflozin Groups Before the Withdrawal of Double-Blind Therapy (Last Value on Double-Blind Treatment)

Characteristics	Placebo (n=2020)	Empagliflozin (n=1961)	P value
Age at baseline, y, mean±SD	70.3±9.7	70.3±9.6	0.993
Women, n (%)	789 (39.1) 770 (39.3)		0.894
Race and ethnicity, n (%)	0.664		
Asian	389 (19.3)	377 (19.2)	
Black or African American	83 (4.1)	76 (3.9)	
White	1424 (70.5)	1406 (71.7)	
Other including mixed race*	122 (6.0)	102 (5.2)	
Geographic region, n (%)			0.499
Asia	367 (18.2)	354 (18.1)	
Europe	906 (44.9)	882 (45.0)	
North America	177 (8.8)	197 (10.0)	
Latin America	472 (23.4)	449 (22.9)	
Estimated glomerular filtration rate, mL·min ⁻¹ ·1.73 m ⁻² , mean±SD	57.7 (20.2)	57.5 (20.0)	0.697
Body weight, kg, mean±SD	80.6 (19.5)	79.6 (19.3)	0.126
Ischemic pathogenesis, n (%)	796 (39.4)	786 (40.1)	0.663
Ejection fraction, n (%)			
≤40%	471 (23.3)	464 (23.7)	0.844
41% to <50%	500 (24.8)	470 (24.0)	
≥50%	1049 (51.9)	1027 (52.4)	
New York Heart Association class I or II, n (%)	1792 (88.7)	1758 (89.6)	0.456
Kansas City Cardiomyopathy Questionnaire, mean±SD	77.1 (19.5)	78.5 (19.7)	0.029
NT-proBNP, pg/mL	1017 (473–1864)	894 (418–1673)	0.002
Systolic blood pressure, mmHg, mean±SD	130.9 (18.3)	129.7 (17.4)	0.038
Serum sodium, mmol/L, mean±SD	140.4 (3.1)	140.6 (3.1)	0.032
Serum potassium, mmol/L, mean±SD	4.49 (0.47)	4.47 (0.50)	0.304
Serum albumin, g/dL, mean±SD	4.36 (0.34)	4.39 (0.33)	0.004
Hemoglobin, g/dL, mean±SD	13.3 (1.7)	14.2 (1.8)	<0.001
Hematocrit, %, mean±SD	40.8 (4.9)	43.6 (5.2)	<0.001
Serum bicarbonate, mmol/L, mean±SD	23.1 (2.8)	22.9 (2.9)	0.009
Serum uric acid, mg/dL, mean±SD	6.5 (1.9)	5.6 (1.7)	<0.001
Hemoglobin A1c, %, mean±SD	6.4 (1.3)	6.3 (1.2)	0.129
Angiotensin converting-enzyme inhibitor or angiotensin receptor blocker, n (%)	1523 (75.4)	1488 (75.9)	0.722
Angiotensin receptor neprilysin inhibitor, n (%)	155 (7.7)	124 (6.3)	0.095
β-Blocker, n (%)	1760 (87.1)	1718 (87.6)	0.649
Mineralocorticoid receptor antagonist, n (%)	934 (46.2)	856 (43.7)	0.101

Values for age, sex, race, geographical region, pathogenesis, and ejection fraction represent values measured before randomization at the start of the trial. All other variables were assessed as the last value on double-blind treatment in those patients with valid data after withdrawal (except hemoglobin A1c, which was not measured after withdrawal). *P* value for NT-proBNP is based on log transformed data. For continuous variables, values are means±SD, except for NT-proBNP where medians and interquartile ranges are shown. NT-proBNP indicates N-terminal prohormone B-type natriuretic peptide.

*Data points were based on patient self-identification.



Figure 1. CONSORT diagram.

Source and disposition of randomized patients included in current analysis. Boldface number refers to total number of patients. The 3981 patients who completed the last-value-on-treatment and withdrawal period were also assessed (with respect to clinical, physiological, and laboratory variables) at baseline, 4 weeks after randomization, and 12 weeks after randomization. These data are shown in Figure 2B, Figure 3, and Table 3. SGLT2 indicates sodium-glucose cotransporter 2.

Physical Examination and Laboratory Assessments

At the start of randomized treatment, compared with placebo, initiation of therapy with empagliflozin was accompanied by decreases in body weight (adjusted mean difference \pm SE, -0.7 ± 0.1 kg), systolic blood pressure $(-1.8\pm0.4 \text{ mm Hg})$, eGFR $(-3.0\pm0.3 \text{ mL·min}^{-1}\cdot1.73)$ m⁻²), NT-proBNP (geometric mean ratio 0.93 [95% Cl, 0.90-0.97]), and serum bicarbonate (-0.4 ± 0.1 mg/dL), all P<0.001. Fasting glucose was not measured at 4 weeks after randomization. At the end of the trial, compared with placebo, the withdrawal of empagliflozin was accompanied by increases in fasting glucose $(+4.0\pm1.3)$ mg/dL), body weight (+0.5 \pm 0.1 kg), systolic blood pressure (+2.3 \pm 0.5 mm Hg), eGFR (+2.7 \pm 0.3 mL·min⁻¹·1.73 m⁻²), NT-proBNP (geometric mean ratio 1.07 [95% Cl, 1.03-1.11]), and serum bicarbonate (+0.3±0.1 mg/ dL), all P<0.01. The magnitude of the effects of empagliflozin during the initiation and withdrawal periods (which were separated by $\approx 1-3$ years) were similar (Table 3 and Figure 2). The pattern of responses was similar in the 2 trials (Table S4).

At the start of the trial, compared with placebo, initiation of therapy with empagliflozin was accompanied by increases in hemoglobin and hematocrit at 4 weeks (adjusted mean difference \pm SE, hemoglobin +0.3 \pm 0.0 g/L and hematocrit +1.1 \pm 0.1%, both *P*<0.001, and these effects became larger after 12 weeks of doubleblind therapy. At the end of the trial, compared with placebo, the withdrawal of empagliflozin was accompanied by significant decreases in hemoglobin (adjusted mean difference \pm SE, -0.4 ± 0.0 g/L) and hematocrit ($-1.5\pm0.1\%$). The magnitude of these changes was similar to those at 4 weeks but meaningfully smaller than those after 12 weeks of randomized treatment (Figure 2). Compared with placebo, uric acid decreased by -1.0 ± 0.0 mg/dL at 4 weeks after the initiation of empagliflozin (P<0.001) and rose by $+0.6\pm0.0$ mg/dL (P<0.001) during the withdrawal period; the magnitude of latter effect was smaller than that observed at the start of randomized treatment (Table 3 and Figure 2). For all changes in physiological and laboratory variables, the pattern of responses was similar in the 2 trials (Table S4).

Although the mean changes during the withdrawal period were often similar to the mean changes during the first 4 weeks of double-blind therapy, there was no meaningful correlation between changes during the 2 time periods in individual patients for any variable, except for a modest correlation for uric acid (ρ –0.26). During the first 4 weeks of double-blind therapy, changes in hematocrit were not meaningfully correlated with changes in NT-proBNP (ρ -0.11 for placebo and -0.21 for empagliflozin) or with changes in body weight (ρ –0.18 for placebo and -0.18 for empagliflozin) but were modestly correlated with changes in serum albumin (p 0.46 for placebo and 0.42 for empagliflozin). Likewise, during the 30-day withdrawal period, changes in hematocrit were not meaningfully correlated with changes in NT-proBNP (ρ -0.20 for placebo and -0.23 for empagliflozin) or

ORIGINAL RESEARCH

Table 2.Cardiovascular Death or Heart FailureHospitalization During the On-Treatment Period (90 DaysFrom Start of Closeout to the End of Double-Blind Treatment) and During the Off-Treatment Withdrawal Period(From End of Double-Blind Treatment to End of WithdrawalPeriod ≈30 Days Later) For Both Trials Combined

Cardiac death or hospitalization	Placebo	Empagliflozin				
Cardiovascular death or hospitalization for heart failure (both trials combined)						
On-treatment (90 days from start of closeout to end of double-blind treatment)						
Number of analyzed patients	3623	3670				
Number of patients with event	163	132				
Time at risk for event, years	1209.7	1229.5				
Incidence rate (events per 100 patient-years at risk)	13.47 (11.48–15.62)	10.74 (8.98–12.64)				
Hazard ratio	0.76 (0.60–0.96)					
Off-treatment (≈30 days from end of double-blind treatment to end of withdrawal period)						
Number of analyzed patients	3381	3418				
Number of patients with event	40	49				
Time at risk for event, y	283.5	287.6				
Incidence rate (events per 100 patient-years at risk)	14.11 (10.08–18.81)	17.04 (12.61–22.13)				
Hazard ratio	1.18 (0.78–1.80)					
Time to first hospitalization for hea	art failure (both trials c	ombined)				
On-treatment (90 days from start of closeout to end of double-blind treatment)						
Number of analyzed patients	3623	3670				
Number of patients with event	107	97				
Time at risk for event, y	1209.7	1229.5				
Incidence rate (events per 100 patient-years at risk)	8.84 (7.25–10.60)	7.89 (6.40–9.53)				
Hazard ratio	0.84 (0.64–1.11)					
Off-treatment (≈30 days from end of double-blind treatment to end of withdrawal period)						
Number of analyzed patients	3381	3418				
Number of patients with event	31	39				
Time at risk for event, y	283.5	287.6				
Incidence rate (events per 100 patient-years at risk)	10.94 (7.43–15.11)	13.56 (9.64–18.14)				
Hazard ratio	1.21 (0.76–1.95)					

Also shown is the principal driving component of the primary end point: time to first heart failure hospitalization. Hazard ratios for EMPEROR Pooled (combined analysis of EMPEROR-Reduced [Empagliflozin Outcome Trial in Patients With Chronic Heart Failure With Reduced Ejection Fraction] and EMPEROR-Preserved [Empagliflozin Outcome Trial in Patients With Chronic Heart Failure With Reduced Ejection Fraction] and EMPEROR-Preserved Ejection Fraction]) estimated from Cox regression models, adjusting for baseline values age, estimated glomerular filtration rate, left ventricular ejection fraction, log troponin T, log N-terminal prohormone B-type natriuretic peptide, diabetes status, sex, region, and study, and use of mineralocorticoid receptor antagonists and angiotensin receptor neprilysin inhibition, as well, at the start of the relevant time period.

with changes in body weight (ρ –0.11 for placebo and –0.18 for empagliflozin) but were modestly correlated with changes in serum albumin (ρ 0.39 for placebo and 0.37 for empagliflozin). These relationships were similar whether patients were withdrawn from placebo or from empagliflozin.

Randomized Withdrawal of SGLT2i in Heart Failure

Additional information on the effects of initiation and withdrawal of treatment are shown in Table 3. Serum albumin increased after the start of empagliflozin and decreased after the withdrawal of the drug, but the changes were very small. Serum sodium and potassium concentrations did not change after 4 weeks of empagliflozin and showed minimal changes after withdrawal of empagliflozin.

DISCUSSION

In a preplanned protocol-specified manner, we withdrew double-blind randomized therapy with placebo or empagliflozin for ≈30 days in ≈7000 patients with heart failure and a preserved or reduced ejection fraction, who in general had been taking the 2 study drugs for periods typically exceeding 1 to 2 years. Before the planned withdrawal, the annualized rate of cardiovascular death or hospitalization for heart failure was lower in the empagliflozin group compared with the placebo group. However, when the study drugs were withdrawn for ≈ 30 days, the rate of cardiovascular death or hospitalization for heart failure increased in the patients who were withdrawn from empagliflozin, but not in those withdrawn from placebo. The hazard ratio for the effect of empagliflozin (versus placebo) changed from 0.76 (95% Cl, 0.60-0.96) while patients were taking their study medications to 1.18 (95% CI, 0.78-1.80) while patients were not taking their study drugs (time period-by-treatment interaction P=0.068; Figure S1). The risk of a major heart failure event was numerically higher after the withdrawal of empagliflozin than in the placebo group before withdrawal (17.0 versus 13.5 events per 100 patient-years). It should be noted that the total number of events during the withdrawal period was <100; thus, the off-treatment observations for major heart failure outcomes should be viewed with considerable caution.

Nevertheless, worsening of clinical status after withdrawal of empagliflozin was evident not only with respect to change in the rate of major heart failure events, but also with respect to changes in health status, which was assessed in \approx 4000 patients. When double-blind therapy was withdrawn, we observed a greater decline in the KCCQ-CSS score within 30 days in the patients who had been withdrawn from treatment with empagliflozin compared with those withdrawn from treatment with placebo (*P*<0.0001). The magnitude of this decline was similar to the magnitude of improvement that had been in these same patients after the first postrandomization assessment of KCCQ-CSS performed 12 weeks after ORIGINAL RESEARCH Article

	4 weeks after initiation of study medications at start of trial (changes from prerandomization baseline value)			${\approx}30$ days after withdrawal of study medications at end of trial (changes from last value on double-blind treatment)		
Variables	Placebo (n=2020)	Empagliflozin (n=1961)	Treatment effect	Placebo (n=2020)	Empagliflozin (n=1961)	Treatment effect
Systolic blood pressure, mm Hg	-0.8±0.3	-2.6±0.3	-1.8±0.4*	-1.1±0.3	+1.3±0.3	+2.3±0.5*
Body weight, kg	+0.1±0.1	-0.5±0.1	-0.7±0.1*	+0.0±0.1	+0.5±0.1	+0.5±0.1*
eGFR, mL·min ⁻¹ ·1.73 m ⁻²	-0.8±0.2	-3.8±0.2	-3.0±0.3*	+0.3±0.2	+3.0±0.2	+2.7±0.3*
NT-proBNP, pg/mL	0.95 (0.92-0.97)	0.88 (0.86-0.91)	0.93* (0.90-0.97)	1.01 (0.98–1.03)	1.08 (1.05–1.10)	1.07* (1.03–1.11)
Serum albumin, g/L	-0.03±0.01	+0.01±0.01	+0.04±0.01*	-0.01±0.01	- 0.05±0.01	-0.05±0.01*
Hemoglobin, g/L	-0.1±0.0	+0.2±0.0	+0.3±0.0*	+0.0±0.0	-0.4±0.0	-0.4±0.0*
Hematocrit, %	-0.6±0.1	+0.6±0.1	+1.1±0.1*	-0.1±0.1	-1.6±0.1	-1.5±0.1*
Serum sodium, mmol/L	0.0±0.1	0.0±0.1	0.0±0.1	-0.1±0.1	-0.1±0.1	-0.1±0.1
Serum potassium, mmol/L	-0.03±0.01	-0.02±0.01	+0.01±0.01	+0.01±0.01	-0.01±0.01	-0.03±0.01‡
Serum bicarbonate, mmol/L	-0.0±0.1	-0.4±0.1	-0.4±0.1*	+0.1±0.1	+0.4±0.1	+0.3±0.1*
Fasting glucose, mg/dL				+0.5±0.9	+4.5±0.9	+4.0±1.3†
Serum uric acid, mg/dL	-0.0±0.0	-1.1±0.0	-1.0±0.0*	-0.0±0.0	+0.5±0.0	+0.6±0.0*
KCCQ-CSS	+3.4±0.3	+4.6±0.3	+1.3±0.4†	-0.4±0.3	-2.0±0.3	-1.6±0.4*

 Table 3.
 Effect of Empagliflozin on Clinical and Laboratory Variables After Initiation of Treatment at the Start of the Trials and

 After Withdrawal of Treatment at the End of the Trials

Results from mixed model for repeated measures model including changes from baseline at week 4, week 12, last value on treatment, and \approx 30 days after treatment discontinuation. Shown are between-group differences for (1) the change from baseline at week 4 for physiological and laboratory variables and from baseline to week 12 for KCCQ; and (2) the change from the value at the end of the withdrawal period compared with last value on treatment. All analyses conducted in a single cohort of patients that provided withdrawal data. Shown are adjusted means \pm SE or adjusted geometric mean ratio (95% CI) for NT-proBNP. (Models for NT-proBNP are based on log transformed data.) The precise number of patients with available data for each physiological or laboratory assessment varied slightly, but, in general, was \approx 1850 to 2000 for each visit in each treatment group. Symbols indicate level of significance for between-group differences: " \mathcal{P} C0.001; \mathcal{H} C0.01; and \mathcal{P} C0.05. eGFR indicates estimated glomerular filtration rate; KCC0-CSS, Kansas City Cardiomyopathy Questionnaire; and NT-proBNP, N-terminal prohormone B-type natriuretic peptide.

the initiation of double-blind therapy, ≈ 1 to 3 years earlier. Although changes in KCCQ-CSS score of 1 to 2 points represent a small difference when assessed in individual patients, these represent a 15% to 30% higher likelihood of a meaningful improvement on a population basis.^{26,27}

Our laboratory assessments can provide insights into the durability of the effects of empagliflozin in the kidney, where it inhibits both SGLT2 and sodium-hydrogen exchanger 3 in the proximal renal tubule.28-30 SGLT2 inhibition lowers blood glucose by promoting glycosuria, and our observation that blood glucose increased after discontinuation of treatment is consistent with a durable effect on urinary glucose excretion. Glycosuria is an important driver of the increase in uric acid excretion and the decrease in serum uric acid during SGLT2 inhibition,³¹ and uric acid increased after withdrawal of empagliflozin. Sodium-hydrogen exchanger 3 inhibition induces a modest bicarbonaturia,29,30 and serum bicarbonate decreases after SGLT2 inhibition.³² We observed a decrease in serum bicarbonate at the start of the trials that persisted for the duration of the trials; this effect was reversed ≈ 30 days after withdrawal of the drug. Last, sodium-hydrogen exchanger 3 inhibition enhances sodium delivery to the macula densa, activating tubuloglomerular feedback and resulting in a decline in glomerular filtration rate.33,34 We observed a decrease in eGFR during the first 4 weeks of randomized therapy, which was fully reversed during the withdrawal period. Taken collectively, these observations suggest that, during long-term treatment with SGLT2 inhibitors, tolerance does not develop to the effect of these drugs to inhibit SGLT2 and sodium-hydrogen exchanger 3 in the proximal tubule.

After discontinuation of empagliflozin, we observed small increases in body weight, systolic blood pressure, and NT-proBNP, which were accompanied by small decreases in hemoglobin, hematocrit, and serum albumin. These findings were the inverse of the changes seen after 4 weeks of double-blind treatment at the start of the trial. These observations might suggest a small increase in total body sodium stores and plasma volume after the withdrawal of therapy. However, increases in weight may have been (in part) related to the cessation of glycosuria, which (by itself) would have yielded a gain of 1.0 to 1.5 kg over the 30-day withdrawal period.³⁵ Furthermore, decreased erythropoietin stimulation and reduced erythrocytosis after the withdrawal of SGLT2 inhibitors may have contributed to the decreases in hemoglobin and hematocrit.^{36,37} We found no or modest correlations between changes in hematocrit and changes in NT-proBNP, changes in body weight or changes in serum albumin during the 30-day withdrawal phase, with similar correlations in both the placebo and empagliflozin groups, suggesting that sodium retention did not selectively occur in one of the treatment groups.³⁸

ORIGINAL RESEARCH



Figure 2. Effect of empagliflozin on clinical efficacy measures during initiation of double-blind therapy and after planned withdrawal of double-blind treatment.

A, The incidence of cardiovascular death or hospitalization for heart failure from 90 days before the start of the closeout period to the end of the double-blind treatment period and during the \approx 30-day withdrawal period. Shown in black are the between-group differences, expressed as hazard ratio (HR) and 95% CIs. Within-group differences for the 2 treatment periods are shown in red for the placebo group and blue for the empagliflozin group. Time-to-event plots for these data are shown in Figure S1. **B**, Values for KCCQ-CSS (±SE) at baseline, 12 weeks after randomization, the last value on double-blind treatment and at the end of the 30-day withdrawal period in the same cohort of 3928 patients who provided KCCQ-CSS data at the end of the planned withdrawal period. *P* values show the between-group difference between empagliflozin versus placebo (1) during initiation of treatment ("on-treatment effect," assessed as between-group difference in the changes from baseline at 12 weeks of randomized treatment) and (2) at the end of the trial ("off-treatment effect," assessed as between-group difference at the end of the withdrawal period compared with last value on double-blind treatment), on the basis of mixed model for repeated measures analyses. KCCQ-CSS indicates Kansas City Cardiomyopathy Questionnaire Clinical Summary Score.

Our findings should be considered in light of certain strengths and limitations. In a large-scale trial, double-blind therapy was withdrawn by protocol design from patients who completed the treatment period, thus minimizing the confounding of previous observational studies.7-11 Our approach was similar to that in another large-scale trial.¹² The group entering the withdrawal phase represented a less impaired subset of our original randomly assigned patients, because these patients were required to have survived and maintained double-blind treatment. Even at the end of the trial, the groups fortunately were well-balanced with respect to assignment to placebo or empagliflozin, because treatment with empagliflozin did not influence the rates of discontinuation of randomized therapy due to death or other causes. Nevertheless, the analyses performed in the patients included in our withdrawal analyses did not have the protection of randomization; however, we made every effort to adjust for all relevant prognostic variables. To address these potential issues and ensure balance between placebo and empagliflozin groups, we would have needed to perform a trial of several thousand patients who had been taking empagliflozin for years and who would be randomly assigned to continue the drug or be withdrawn into treatment with placebo for a period of several months or longer. Such a design has been carried in the past to evaluate the longterm efficacy of digoxin.¹³

Furthermore, it should be emphasized that the withdrawal of therapy was abrupt (as per protocol) and the period of follow-up off-treatment was brief (≈30 days). The proximal tubular effect of SGLT2 inhibitors elicits compensatory downstream antinatriuretic and antiaquaretic mechanisms that truncate any expected changes in urinary sodium and water excretion.15-19 Abrupt withdrawal allows these counterregulatory mechanisms to become apparent as rebound phenomena, which have been reported when diuretics and vasodilator drugs are used for the treatment of heart failure.³⁹⁻⁴¹ Given the numerically higher event rate after the withdrawal of empagliflozin than in the placebo group before withdrawal, it is reasonable to postulate that a short-term rebound effect may have amplified the magnitude of the changes that we observed after the withdrawal of empagliflozin,42 thus enhancing estimates of an on-treatment

ORIGINAL RESEARCH Article



Figure 3. Effect of empagliflozin on physiological and laboratory assessments during initiation of double-blind therapy and after planned withdrawal of double-blind treatment.

Values at baseline (BL), 4 and 12 weeks after randomization, the last value on double-blind treatment (LVOT), and at the end of the 30-day withdrawal period (W) in the cohort of 3981 patients who provided data during the withdrawal period. Shown are adjusted mean changes from baseline±SE for all variables except for NT-proBNP, for which adjusted geometric mean ratio (95% CI) is displayed. *P* values on the left side of each graft represent the significance of between-group changes at 4 weeks from baseline, and *P* values on the right side of each graft represent the significance of between-group changes at the end of withdrawal compared with LVOT, on the basis of mixed model for repeated (*Continued*)

Figure 3 Continued. measures analyses. For NT-proBNP the analyses were performed on log-transformed values. The precise number of patients with available data for each physiological and laboratory assessment varied slightly but, in general, was ≈1850 to 2000 for each visit in each treatment group. BP indicates blood pressure; eGFR, estimated glomerular filtration rate; and NT-proBNP, N-terminal prohormone B-type natriuretic peptide.

effect. Conversely, our ability to assess the full effect of the discontinuation of treatment on clinical events may have been limited by the brevity of our withdrawal period, which was shorter than that in other drug withdrawal trials in patients with heart failure.^{14,43,44} With additional time off-treatment, we would have observed more events, thus allowing us to define the between-group differences with greater certainty.

In conclusion, in 2 large-scale trials in which we withdrew randomized double-blind treatment with placebo or empagliflozin in patients with heart failure with a reduced or preserved ejection fraction, we demonstrated a persistent clinical benefit of empagliflozin even after years of treatment, which dissipated after withdrawal. Maintenance of the clinical effect of the drug was accompanied by persistence of the physiological and laboratory changes that are characteristic of SGLT2 inhibition. These findings indicate that tolerance does not develop during long-term treatment with empagliflozin in patients with heart failure and that cessation of treatment even for short periods of time may have deleterious consequences.

ARTICLE INFORMATION

Received May 26, 2023; accepted August 3, 2023.

Affiliations

Baylor University Medical Center, Dallas, TX (M.P.). Imperial College, London, United Kingdom (M.P.). Baylor Scott and White Research Institute, Dallas, TX (J.B.). Department of Medicine, University of Mississippi School of Medicine, Jackson, MS (J.B.). Boehringer Ingelheim Pharma GmbH and Co KG, Biberach, Germany (C.Z.). Department of Medical Statistics, London School of Hygiene & Tropical Medicine, United Kingdom (S.J.P.). Boehringer Ingelheim International GmbH, Ingelheim, Germany (M.B.). First Department of Medicine, Faculty of Medicine Mannheim, University of Heidelberg, Germany (M.B.). Cardiovascular Research and Development Center, Faculty of Medicine of the University of Porto, Portugal (J.P.F.). Centre d'Investigations Cliniques Plurithématique 14-33, and Inserm U1116, CHRU, F-CRIN INI-CRCT (Cardiovascular and Renal Clinical Trialists), Université de Lorraine, Nancy, France (J.P.F., F.Z.). National and Kapodistrian University of Athens School of Medicine, Athens University Hospital Attikon, Athens, Greece (G.F.). Division of Cardiology, University of Texas Southwestern Medical Center, Dallas (M.S.U.). Department of Cardiology (CVK); and Berlin Institute of Health Center for Regenerative Therapies (BCRT); German Centre for Cardiovascular Research (DZHK) partner site Berlin; Charité Universitätsmedizin Berlin (S.D.A.).

Sources of Funding

This work was funded by Boehringer Ingelheim and Eli Lilly & Company.

Disclosures

Dr Butler reports consulting fees from Abbott Fund, American Regent, Amgen, Applied Therapeutics, Array Biopharma, AstraZeneca, Bayer, Boehringer Ingelheim, Cardiac Dimensions, Cardior, CVRx, Inc, Cytokinetics, Edwards Lifesciences, Element Sciences, Eli Lilly & Company, Impulse Dynamics, Imbria, Inventiva, Innolife, Janssen Global, Lexicon Pharmaceuticals, Liva Nova USA, Luitpold Pharmaceuticals, Medtronic, Merck, Novartis, Novo Nordisk, Pharmacosmos, Roche Diagnostics, Occlutech, Relypsa, SQ Innovation, Sequana, Stelthpeptide, Vifor Pharma. Dr Zannad reports consulting fees from AstraZeneca, Bayer, Boehringer Ingelheim, Cellprothera, Merck, Novartis, Novo Nordisk, Owkin, Pfizer, Servier Affaires Medicale, Vifor Fresenius. Dr Ferreira reports consulting fees from Boehringer Ingelheim. Dr Packer reports consulting fees from 89bio, Abbvie, Altimmune, Alnylam, Amarin, Amgen, Ardelyx, AstraZeneca, Boehringer Ingelheim, Caladrius, Casana, CSL Behring, Cytokinetics, Imara, Lilly, Medtronic, Moderna, Novartis, Reata, Relypsa, Salamandra. Dr Anker reports consulting fees from Amgen, AstraZeneca, Bayer, Boehringer Ingelheim, Brahms GmbH, Cardiac Dimensions, Cardior, Cordio, CVRx, Inc., Edwards Lifesciences, GlaxoSmithKline, Impulse Dynamics (USA) Inc., Novartis, Pfizer, Servier, V-Wave, Vectorious, Vifor International. Dr Pocock reports consulting fees from Boehringer Ingelheim. Drs Brueckmann and Zeller are employees of Boehringer Ingelheim. Dr Usman reports no conflicts.

Supplemental Material

Figure S1 Tables S1-S4

REFERENCES

- Vaduganathan M, Claggett BL, Inciardi RM, Fonarow GC, McMurray JJV, Solomon SD. Estimating the benefits of combination medical therapy in heart failure with mildly reduced and preserved ejection fraction. *Circulation*. 2022;145:1741–1743. doi: 10.1161/CIRCULATIONAHA.121.058929
- Vaduganathan M, Docherty KF, Claggett BL, Jhund PS, de Boer RA, Hernandez AF, Inzucchi SE, Kosiborod MN, Lam CSP, Martinez F, et al. SGLT-2 inhibitors in patients with heart failure: a comprehensive meta-analysis of five randomised controlled trials. *Lancet.* 2022;400:757–767. doi: 10.1016/S0140-6736(22)01429-5
- Packer M, Medina N, Yushak M. Role of the renin-angiotensin system in the development of hemodynamic and clinical tolerance to long-term prazosin therapy in patients with severe chronic heart failure. J Am Coll Cardiol. 1986;7:671–680. doi: 10.1016/s0735-1097(86)80479-x
- Colucci WS, Alexander RW, Williams GH, Rude RE, Holman BL, Konstam MA, Wynne J, Mudge GH Jr, Braunwald E. Decreased lymphocyte betaadrenergic-receptor density in patients with heart failure and tolerance to the beta-adrenergic agonist pirbuterol. *N Engl J Med.* 1981;305:185–190. doi: 10.1056/NEJM198107233050402
- Packer M, Lee WH, Kessler PD, Gottlieb SS, Medina N, Yushak M. Prevention and reversal of nitrate tolerance in patients with congestive heart failure. N Engl J Med. 1987;317:799–804. doi: 10.1056/NEJM198709243171304
- Packer M, Meller J, Medina N, Yushak M, Gorlin R. Hemodynamic characterization of tolerance to long-term hydralazine therapy in severe chronic heart failure. N Engl J Med. 1982;306:57–62. doi: 10.1056/NEJM198201143060201
- Fonarow GC, Abraham WT, Albert NM, Stough WG, Gheorghiade M, Greenberg BH, O'Connor CM, Sun JL, Yancy CW, Young JB; OPTIMIZE-HF Investigators and Coordinators. Influence of beta-blocker continuation or withdrawal on outcomes in patients hospitalized with heart failure: findings from the OPTIMIZE-HF program. *J Am Coll Cardiol*. 2008;52:190–199. doi: 10.1016/j.jacc.2008.03.048
- Butler J, Young JB, Abraham WT, Bourge RC, Adams KF Jr, Clare R, O'Connor C; ESCAPE Investigators. Beta-blocker use and outcomes among hospitalized heart failure patients. *J Am Coll Cardiol*. 2006;47:2462– 2469. doi: 10.1016/j.jacc.2006.03.030
- Chen Y, Qiu Z, Jiang J, Su X, Huang F, Tang J, Jin W. Outcomes of spironolactone withdrawal in dilated cardiomyopathy with improved ejection fraction. *Front Cardiovasc Med.* 2021;8:1080.
- Gilstrap LG, Fonarow GC, Desai AS, Liang L, Matsouaka R, DeVore AD, Smith EE, Heidenreich P, Hernandez AF, Yancy CW, et al. Initiation, continuation, or withdrawal of angiotensin-converting enzyme inhibitors/angiotensin receptor blockers and outcomes in patients hospitalized with heart failure with reduced ejection fraction. *J Am Heart Assoc.* 2017;6:e004675. doi: 10.1161/JAHA.116.004675
- Amin OA, Alaarag AF. The impact of discontinuation of sacubitril-valsartan and shifting to angiotensin-converting enzyme inhibitor or angiotensin receptor blocker in patients with heart failure with reduced ejection fraction. *Anatol J Cardiol.* 2021;25:163–169. doi: 10.14744/AnatolJCardiol.2020.39267

- Packer M, Pitt B, Rouleau JL, Swedberg K, DeMets DL, Fisher L. Long-term effects of flosequinan on the morbidity and mortality of patients with severe chronic heart failure: primary results of the PROFILE trial after 24 years. *JACC Heart Fail*. 2017;5:399–407. doi: 10.1016/j.jchf.2017.03.003
- The Digitalis Investigation Group. The effect of digoxin on mortality and morbidity in patients with heart failure. N Engl J Med. 1997;336:525–533.
- Halliday BP, Wassall R, Lota AS, Khalique Z, Gregson J, Newsome S, Jackson R, Rahneva T, Wage R, Smith G, et al. Withdrawal of pharmacological treatment for heart failure in patients with recovered dilated cardiomyopathy (TRED-HF): an open-label, pilot, randomised trial. *Lancet* 2019;393:61–73. doi: 10.1016/S0140-6736(18)32484-X
- Boorsma EM, Beusekamp JC, Ter Maaten JM, Figarska SM, Danser AHJ, van Veldhuisen DJ, van der Meer P, Heerspink HJL, Damman K, Voors AA. Effects of empagliflozin on renal sodium and glucose handling in patients with acute heart failure. *Eur J Heart Fail*. 2021;23:68–78. doi: 10.1002/ejhf.2066
- 16. Scholtes RA, Muskiet MHA, van Baar MJB, Hesp AC, Greasley PJ, Hammarstedt A, Karlsson C, Hallow KM, Danser AHJ, Heerspink HJL, et al. The adaptive renal response for volume homeostasis during 2 weeks of dapagliflozin treatment in people with type 2 diabetes and preserved renal function on a sodium-controlled diet. *Kidney Int Rep.* 2022;7:1084–1092. doi: 10.1016/j.ekir.2022.02.023
- Scholtes RA, Muskiet MHA, van Baar MJB, Hesp AC, Greasley PJ, Karlsson C, Hammarstedt A, Arya N, van Raalte DH, Heerspink HJL. Natriuretic effect of two weeks of dapagliflozin treatment in patients with type 2 diabetes and preserved kidney function during standardized sodium intake: results of the DAPASALT trial. *Diabetes Care.* 2021;44:440–447. doi: 10.2337/dc20-2604
- van Ruiten CC, Smits MM, Kok MD, Serné EH, van Raalte DH, Kramer MHH, Nieuwdorp M, IJzerman RG. Mechanisms underlying the blood pressure lowering effects of dapagliflozin, exenatide, and their combination in people with type 2 diabetes: a secondary analysis of a randomized trial. *Cardiovasc Diabetol*. 2022;21:63. doi: 10.1186/s12933-022-01492-x
- Tanaka H, Takano K, lijima H, Kubo H, Maruyama N, Hashimoto T, Arakawa K, Togo M, Inagaki N, Kaku K. Factors affecting canagliflozin-induced transient urine volume increase in patients with type 2 diabetes mellitus. *Adv Ther.* 2017;34:436–451. doi: 10.1007/s12325-016-0457-8
- Heise T, Jordan J, Wanner C, Heer M, Macha S, Mattheus M, Lund SS, Woerle HJ, Broedl UC. Acute pharmacodynamic effects of empagliflozin with and without diuretic agents in patients with type 2 diabetes mellitus. *Clin Ther.* 2016;38:2248–2264.e5. doi: 10.1016/j.clinthera.2016.08.008
- Anker SD, Butler J, Filippatos G, Ferreira JP, Bocchi E, Böhm M, Brunner-La Rocca HP, Choi DJ, Chopra V, Chuquiure-Valenzuela E, et al; EMPEROR-Preserved Trial Investigators. Empagliflozin in heart failure with a preserved ejection fraction. N Engl J Med. 2021;385:1451–1461. doi: 10.1056/NEJMoa2107038
- Packer M, Anker SD, Butler J, Filippatos G, Pocock SJ, Carson P, Januzzi J, Verma S, Tsutsui H, Brueckmann M, et al; EMPEROR-Reduced Trial Investigators. Cardiovascular and renal outcomes with empagliflozin in heart failure. *N Engl J Med.* 2020;383:1413–1424. doi: 10.1056/NEJMoa2022190
- Packer M, Butler J, Filippatos G, Zannad F, Ferreira JP, Zeller C, Brueckmann M, Jamal W, Pocock SJ, Anker SD; EMPEROR Trial Committees and Investigators. Design of a prospective patient-level pooled analysis of two parallel trials of empagliflozin in patients with established heart failure. *Eur J Heart Fail*. 2020;22:2393–2398. doi: 10.1002/ejhf.2065
- Pocock SJ, Ferreira JP, Gregson J, Anker SD, Butler J, Filippatos G, Gollop ND, Iwata T, Brueckmann M, Januzzi JL, et al. Novel biomarker-driven prognostic models to predict morbidity and mortality in chronic heart failure: the EMPEROR-Reduced trial. *Eur Heart J.* 2021;42:4455–4464. doi: 10.1093/eurheartj/ehab579
- Pocock SJ, Ferreira JP, Packer M, Zannad F, Filippatos G, Kondo T, McMurray JJV, Solomon SD, Januzzi JL, Iwata T, et al. Biomarker-driven prognostic models in chronic heart failure with preserved ejection fraction: the EMPEROR-Preserved trial. *Eur J Heart Fail*. 2022;24:1869–1878. doi: 10.1002/ejhf.2607
- Butler J, Anker SD, Filippatos G, Khan MS, Ferreira JP, Pocock SJ, Giannetti N, Januzzi JL, Piña IL, Lam CSP, et al; EMPEROR-Reduced Trial Committees and Investigators. Empagliflozin and health-related quality of life outcomes in patients with heart failure with reduced ejection fraction: the EMPEROR-Reduced trial. *Eur Heart J.* 2021;42:1203–1212. doi: 10.1093/eurheartj/ehaa1007
- Butler J, Filippatos G, Siddiqi TJ, Brueckmann M, Böhm M, Chopra VK, Ferreira JP, Januzzi JL, Kaul S, Piña IL, et al. Empagliflozin, health status,

and quality of life in patients with heart failure and preserved ejection fraction: the EMPEROR-Preserved trial. *Circulation*. 2022;145:184–193.

- Borges-Júnior FA, Silva Dos Santos D, Benetti A, Polidoro JZ, Wisnivesky ACT, Crajoinas RO, Antônio EL, Jensen L, Caramelli B, Malnic G, et al. Empagliflozin inhibits proximal tubule NHE3 activity, preserves GFR, and restores euvolemia in nondiabetic rats with induced heart failure. *J Am Soc Nephrol*. 2021;32:1616–1629.
- Onishi A, Fu Y, Patel R, Darshi M, Crespo-Masip M, Huang W, Song P, Freeman B, Kim YC, Soleimani M, et al. A role for tubular Na+/H+ exchanger NHE3 in the natriuretic effect of the SGLT2 inhibitor empagliflozin. *Am J Physiol Renal Physiol.* 2020;319:F712–F728. doi: 10.1152/ajprenal.00264.2020
- Pessoa TD, Campos LC, Carraro-Lacroix L, Girardi AC, Malnic G. Functional role of glucose metabolism, osmotic stress, and sodium-glucose cotransporter isoform-mediated transport on Na+/H+ exchanger isoform 3 activity in the renal proximal tubule. *J Am Soc Nephrol.* 2014;25:2028–2039. doi: 10.1681/ASN.2013060588
- Novikov A, Fu Y, Huang W, Freeman B, Patel R, van Ginkel C, Koepsell H, Busslinger M, Onishi A, Nespoux J, et al. SGLT2 inhibition and renal urate excretion: role of luminal glucose, GLUT9, and URAT1. *Am J Physiol Renal Physiol*. 2019;316:F173–F185. doi: 10.1152/ajprenal.00462.2018
- Segar MW, Kolkailah AA, Frederich R, Pong A, Cannon CP, Cosentino F, Dagogo-Jack S, McGuire DK, Pratley RE, Liu CC, et al. Mediators of ertugliflozin effects on heart failure and kidney outcomes among patients with type 2 diabetes mellitus. *Diabetes Obes Metab.* 2022;24:1829–1839. doi: 10.1111/dom.14769
- Vallon V, Thomson SC. The tubular hypothesis of nephron filtration and diabetic kidney disease. Nat Rev Nephrol. 2020;16:317–336. doi: 10.1038/s41581-020-0256-y
- Poursharif S, Hamza S, Braam B. Changes in proximal tubular reabsorption modulate microvascular regulation via the TGF system. *Int J Mol Sci*. 2022;23:11203. doi: 10.3390/ijms231911203
- 35. Op den Kamp YJM, de Ligt M, Dautzenberg B, Kornips E, Esterline R, Hesselink MKC, Hoeks J, Schrauwen-Hinderling VB, Havekes B, Oscarsson J, et al. Effects of the SGLT2 inhibitor dapagliflozin on energy metabolism in patients with type 2 diabetes: a randomized, double-blind crossover trial. *Diabetes Care*. 2021;44:1334–1343. doi: 10.2337/dc20-2887
- Yamada T, Sakaguchi K, Okada Y, Miura H, Otowa-Suematsu N, So A, Komada H, Hirota Y, Ohara T, Kuroki Y, et al. Analysis of time-dependent alterations of parameters related to erythrocytes after ipragliflozin initiation. *Diabetol Int.* 2020;12:197–206. doi: 10.1007/s13340-020-00474-2
- 37. Fuchs Andersen C, Omar M, Glenthøj A, El Fassi D, Møller HJ, Lindholm Kurtzhals JA, Styrishave B, Kistorp C, Tuxen C, Poulsen MK, et al. Effects of empagliflozin on erythropoiesis in heart failure: data from the Empire HF trial. *Eur J Heart Fail.* 2022;25:226–234. doi: 10.1002/ejhf.2735
- Packer M, Anker SD, Butler J, Filippatos G, Ferreira JP, Pocock SJ, Carson P, Anand I, Doehner W, Haass M, et al. Effect of empagliflozin on the clinical stability of patients with heart failure and a reduced ejection fraction: the EMPEROR-Reduced trial. *Circulation*. 2021;143:326–336. doi: 10.1161/CIRCULATIONAHA.120.051783
- Packer M, Meller J, Medina N, Gorlin R, Herman MV. Rebound hemodynamic events after the abrupt withdrawal of nitroprusside in patients with severe chronic heart failure. N Engl J Med. 1979;301:1193–1197. doi: 10.1056/NEJM197911293012201
- Missouris CG, MacGregor GA. Rebound sodium and water retention occurs when diuretic treatment is stopped. *BMJ.* 1998;316:628. doi: 10.1136/bmj.316.7131.628a
- Hollenberg NK. Set point for sodium homeostasis: surfeit, deficit, and their implications. *Kidney Int*. 1980;17:423–429. doi: 10.1038/ki.1980.50
- Packer M, Meller J, Medina N, Yushak M, Gorlin R. Determinants of drug response in severe chronic heart failure. 1. Activation of vasoconstrictor forces during vasodilator therapy. *Circulation*. 1981;64:506–514. doi: 10.1161/01.cir.64.3.506
- Packer M, Gheorghiade M, Young JB, Costantini PJ, Adams KF, Cody RJ, Smith LK, Van Voorhees L, Gourley LA, Jolly MK. Withdrawal of digoxin from patients with chronic heart failure treated with angiotensin-convertingenzyme inhibitors. RADIANCE Study. N Engl J Med. 1993;329:1–7. doi: 10.1056/NEJM199307013290101
- Pflugfelder PW, Baird MG, Tonkon MJ, DiBianco R, Pitt B. Clinical consequences of angiotensin-converting enzyme inhibitor withdrawal in chronic heart failure: a double-blind, placebo-controlled study of quinapril. The Quinapril Heart Failure Trial Investigators. *J Am Coll Cardiol*. 1993;22:1557– 1563. doi: 10.1016/0735-1097(93)90578-0

Downloaded from http://ahajournals.org by on January 9, 2024